Author's response to reviews

Title: Outcome measurement in clinical trials for Ulcerative Colitis: toward standardisation

Authors:

Rachel M Cooney (rachel.cooney@clinical-medicine.oxford.ac.uk)
Bryan F Warren (bryan.warren@orh.nhs.uk)
Douglas G Altman (doug.altman@cancer.org.uk)
Maria T Abreu (maria.abreu@mssm.edu)
Simon PL Travis (simon.travis@ndm.ox.ac.uk)

Version: 6 Date: 18 June 2007

Author's response to reviews: see over
full stop removed from the title.

Author list put on one line. Corresponding author indicated in this list with a symbol e.g. *

Website addresses removed from the text completely

Competing interests, Authors' contributions and acknowledgements Put between the Conclusions and references sections, and in that order.

Blank page removed.

Tables section given heading: Tables

I have mentioned the registration number of the trial that is discussed briefly in the article EUDRACT no: 2004-004077-29. As this paper is not a report of a trial I have not put it in the abstract.

Table 6 changed to text box 1 as this more accurately describes it, appropriate changes made to body of text.

Full stops added where missing in references and references 14 and 15 reformatted.